Author's response to reviews

Title: The association of body image distortion with weight control behaviors, diet behaviors, physical activity, sadness, and suicidal ideation among Korean high school students: a cross-sectional study

Authors:
Jounghee Lee (jougheelee@kgu.ac.kr)
Young Min Lee (ymlee@swu.ac.kr)

Version: 1 Date: 09 Dec 2015

Author's response to reviews:

Responses from the authors are incorporated in blue text. Changes are shown as blue text in the tables in Author's response document and highlighted in yellow in the manuscript.